Author's response to reviews

Title: An unusual presentation of multiple myeloma: a case report

Authors:

Catherine B Molloy (catherine_molloy@doctors.org.uk)
Rahul Peck (rahul.peck@btopenworld.com)
Stephen J Bonny (sjbonny@doctors.net.uk)
Simon N Jowitt (simon.jowitt@stockport-tr.nwest.nhs.uk)
John Denton (john.denton@manchester.ac.uk)
Anthony J Freemont (tony.freemont@manchester.ac.uk)
Abbas A Ismail (abbas.ismail@stockport-tr.nwest.nhs.uk)

Version: 5 Date: 30 August 2007

Author's response to reviews:

Dear BioMed Central Editorial Production team,
I have made the following changes to my manuscript.
1. I have aligned the title to the left
2. titles have been removed,
3. author email addresses have been included,
4. I've removed the conflicts of interest from the title page.
5. I added a background and case presentation
6. the authors have all read the final manuscript
7. The figure has been removed from the written part of the manuscript and a figure legends section created.

Thank you, yours sincerely, Dr Stephen Bonny